\begin{thebibliography}{10}

\bibitem{Bettstetter:03}
Christian Bettstetter, Giovanni Resta, and Paolo Santi.
\newblock {The node distribution of the random waypoint mobility model for
  wireless ad hoc networks}.
\newblock {\em IEEE Transactions on Mobile Computing}, 2(3):257--269, Jul-Sep
  2003.

\bibitem{Bokareva:04}
Tatiana Bokareva, Nirupama Bulusu, and Sanjay Jha.
\newblock A performance comparison of data dissemination protocols for wireless
  sensor networks.
\newblock In {\em Proceedings of the Forty Seventh Annual IEEE Global
  Telecommunications Conference (GLOBECOM 2004) Ad hoc and Wireless Sensor
  Networks Track}, Dallas, TX, 2004.

\bibitem{he03speed}
T.~He, J.~Stankovic, C.~Lu, and T.~Abdelzaher.
\newblock Speed: A stateless protocol for real-time communication in sensor
  networks.
\newblock In {\em Proceedings of the International Conference on Distributed
  Computing Systems (ICDCS 2003.}, 2003.

\bibitem{AIDA:04}
Tian He, Brian~M. Blum, John~A. Stankovic, and Tarek Abdelzaher.
\newblock Aida: Adaptive application-independent data aggregation in wireless
  sensor networks.
\newblock {\em Trans. on Embedded Computing Sys.}, 3(2):426--457, 2004.

\bibitem{Estrin:03}
John Heidemann, Fabio Silva, and Deborah Estrin.
\newblock Matching data dissemination algorithms to application requirements.
\newblock In {\em Proceedings of ACM SenSys 2003}, Los Angeles, CA, November
  2003.

\bibitem{Heinzelman:99}
Wendi~Rabiner Heinzelman, Joanna Kulik, and Hari Balakrishnan.
\newblock Adaptive protocols for information dissemination in wireless sensor
  networks.
\newblock In {\em MobiCom '99: Proceedings of the 5th annual ACM/IEEE
  international conference on Mobile computing and networking}, pages 174--185,
  New York, NY, USA, 1999. ACM Press.

\bibitem{Hill2000}
Jason Hill, Robert Szewczyk, Alec Woo, Seth Hollar, David Culler, and Kristofer
  Pister.
\newblock System architecture directions for networked sensors.
\newblock In {\em Proceedings of the ninth international conference on
  Architectural support for programming languages and operating systems}, pages
  93--104. ACM Press, 2000.

\bibitem{hull2004mitigating}
Bret Hull, Kyle Jamieson, and Hari Balakrishnan.
\newblock {Mitigating Congestion in Wireless Sensor Networks}.
\newblock In {\em ACM SenSys 2004}, Baltimore, MD, November 2004.

\bibitem{Estrin_DD:00}
Chalermek Intanagonwiwat, Ramesh Govindan, and Deborah Estrin.
\newblock Directed diffusion: A scalable and robust communication paradigm for
  sensor networks.
\newblock In {\em Proceedings of the Sixth Annual ACM/IEEE International
  Conference on Mobile Computing and Networking (MobiCom 2000)}, pages 56--67,
  Boston, MA, 2000.

\bibitem{Johnson_RWP:96}
David~B Johnson and David~A Maltz.
\newblock Dynamic source routing in ad hoc wireless networks.
\newblock In Imielinski and Korth, editors, {\em Mobile Computing}, volume 353.
  Kluwer Academic Publishers, 1996.

\bibitem{GPSR:00}
B.~Karp and H.T. Kung.
\newblock Greedy perimeter stateless routing for wireless networks.
\newblock In {\em In Proceedings of the Sixth Annual ACM/IEEE International
  Conference on Mobile Computing and Networking (MobiCom 2000)}, Boston, MA,
  2000.

\bibitem{khanna:04}
Gunjan Khanna, Saurabh Bagchi, and Yu-Sung Wu.
\newblock Fault tolerant energy aware data dissemination protocol in sensor
  network.
\newblock In {\em Proceedings of the IEEE Dependable Systems and Networks
  Conference (DSN 2004)}, Florence, Italy, June 2004.

\bibitem{GPS:01}
US~Naval Observatory (USNO)~GPS Operations.
\newblock Global positioning system.
\newblock http://tycho.usno.navy.mil/gps.html, April 2001.

\bibitem{Pottie:00}
G.~J. Pottie and W.~J. Kaiser.
\newblock Wireless integrated network sensors.
\newblock {\em Commun. ACM}, 43(5):51--58, 2000.

\bibitem{GHT:02}
S.~Ratnasamy, B.~Karp, L.~Yin, F.~Yu, D.~Estrin, R.~Govindan, and S.~Shenker.
\newblock Ght: A geographic hash table for data-centric storage in sensornets.
\newblock In {\em Proceedings of the First ACM International Workshop on
  Wireless Sensor Networks and Applications (WSNA)}, Atlanta, Georgia,
  September 2002.

\bibitem{Druschel:01}
A.~Rowstron and P.~Druschel.
\newblock Pastry: Scalable, distributed object location and routing for
  large-scale peer-to-peer systems.
\newblock In {\em Proceedings of the 18th IFIP/ACM Symposium on Operating
  Systems Principles}, August 2001.

\bibitem{NS-2}
{The VINT Project}.
\newblock {Network Simulator - NS-2}.
\newblock \url{http://www.isi.edu/nsnam/ns}.

\bibitem{Zhang_TTDD:02}
Fan Ye, Haiyun Luo, Jerry Cheng, Songwu Lu, and Lixia Zhang.
\newblock Two-tier data dissemination model for large-scale wireless sensors
  networks.
\newblock In {\em Proceedings of the Eight Annual ACM/IEEE International
  Conference on Mobile Computing and Networking (MobiCom 2002)}, pages
  148--159, Atlanta, GA, 2002.

\bibitem{RB:03}
Wensheng Zhang, Guohong Cao, and Tom~La Porta.
\newblock Data dissemination with ring-based index for wireless sensor
  networks.
\newblock In {\em Proceedings of Eleventh IEEE International Conference on
  Network Protocols (ICNP 2003)}, Atlanta, GA, 2003.

\end{thebibliography}

% $Id: relwork.tex,v 1.5 2005/04/08 21:21:45 gokhale Exp $

\section{Related Work}
\label{sec:relwork}

Sensor networks have been investigated actively over the past few 
years. SPIN~\cite{Heinzelman:99} is a family of protocols used for efficiently
disseminating data over a sensor network. It uses data negotiation
and resource-adaptive algorithms to reduce energy consumption.
Nodes assign a high-level name ({\em meta-data}) to their data
and perform meta-data negotations before data communication to
ensure redundant data is not sent.

GHT~\cite{GHT:02} is a data-centric storage based dissemination scheme
which uses hash function to determine nodes where data are kept.
However, GHT requires data be pushed to these nodes such that
sink nodes can access it. It may thus introduce data transfers
even when sink actually is not querying.

In Directed diffusion (DD)~\cite{Estrin_DD:00}
a sink broadcasts {\em interest} (data request) message throughout
the network. When source receives an interest message it sends
data in a similar fashion. On receiving the data response (
possibly from multiple neighbors) one path is reinforced and is
used for further communication. Multiple paths may be used by 
a sink node for redundancy.

Two-tier Data Dissemination (TTDD)~\cite{Zhang_TTDD:02} uses a
two-tier architecture to disseminate the data to interested nodes. A
grid is formed with the source node at one of the crossing points of
the grid. Thus, the sensing space is divided by the grid into small
cells. Only the nodes at the crossing 
points (called {\em dissemination nodes}) can forward the data. A sink
node, which is interested in receiving a specific data, floods its
local cell in order to discover the closest dissemination node. Each
dissemination node contains forwarding information for each request of
data it receives. On discovery of a dissemination point, the data request is
forwarded to source node by the dissemination nodes, followed by data
report being sent to the sink. Here, the smaller cell is the first
tier, while the higher tier is made of the dissemination nodes at the
grid points, which makes the two tier architecture. 

Similar to TTDD EASM employs a virtual grid, however EASM uses this
grid structure simply to find out the grid center location to be filled
in sink queries while TTDD uses it for localizing sink mobility
impact.

Ring-based index~\cite{RB:03} is a rendezvous based data dissemination
scheme.  A rendezvous point is used as a node to communicate between
sink and source nodes. 
An index ring is used for replication, where nodes are {\em forwarding 
nodes} which forward the data requests to next clockwise neighbor
in the ring and {\em index nodes} which know the location of and 
hence can forward the request to source nodes.
EASM is similar to ring-based index in that it also uses rendezvous
node to query for source node location. However, the way replication
is handled is fundamentally different. Instead of a ring, EASM uses
{\em n} depth replication to elect {\em forwarders} and {\em
interceptors}.  Furthermore, EASM handles local as well as global sink
mobility patterns while ring-based is suited for local sink movements.

Most of the already proposed schemes, which rely on broadcast
(of interest or response messages) may not work well in large
networks, with high number of sink-source pairs, especially when
sinks exhibit high degrees of mobility. Data requests and responses
are always sent to a location in the sensing space, thereby resulting
in significant energy savings. Node replication reduces hot-spot
message complexity and increases fault tolerance. Finally it is able
to handle high sink speeds with marginal increase in energy
consumption. 

%%% Local Variables: 
%%% mode: latex
%%% TeX-master: "sensys05"
%%% End: 
